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Figure 4. Experimental factors for 3D surface plots of tapioca powder conversion to fluorescent
carbon dots.
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3.3. Evaluation Performance between Artificial Neural Network (ANN) and Response Surface Methodology
(RSM) on the Yield of Photoluminescent Quantum Yield

Artificial neural Network (ANN) is a system that mimics the naturally inspired computational
model. Thus, it emulates the workings of the human brain to take in certain connections among
information inputs and yield outputs through trained data [27-30].

From Table 7 below, it shows the relationship between the response surface methodology and
the artificial neural network performance of the trained data (see Figure 5). The best output for
photoluminescent quantum yield was obtained at No. 11 at actual experimental value of 27.75%,
RSM predicted value of 27.38%, and ANN predicted value of 26.25%. The training of the data set
was conducted by Matlab R2015a (8.5.0.197613), utilizing Lavenberg—-Marquardt algorithm (LMA).
The LMA is based on the training neural network through iteration and reiteration of data set weight
and bias values as shown on Table 8 [31-33].

Table 7. Response surface methodology and artificial neural network.

Factor-A Factor-B  Factor-C Factor-D Solvent Exp. Actual RSM  ANN.

Osrf;ider Temperature Dossage Time (mL) Value Pred.  Pred.
Q) (gram) (min) (H,O/C3H¢O/NaOH) (PLQY%) Value Value

1 75 0.10 45 8.00 14.67 1441 1246
2 175 0.10 45 8.00 21.05 20.89  21.31
3 75 0.50 45 8.00 22.80 2235 2274
4 175 0.50 45 8.00 19.96 1913 19.93
5 75 0.10 105 8.00 14.00 13.01  14.76
6 175 0.10 105 8.00 25.27 2613 24.84
7 75 0.50 105 8.00 20.15 2152 19.99
8 175 0.50 105 8.00 24.87 2494 2496
9 75 0.10 45 40.00 24.82 2439 2451
10 175 0.10 45 40.00 20.99 19.88  21.10
11 170 0.1 100 12.00 27.75 2738 2625
12 175 0.50 45 40.00 12.53 13.16  15.80
13 75 0.10 105 40.00 17.90 1899  17.56
14 175 0.10 105 40.00 21.04 2113 21.07
15 75 0.50 105 40.00 22.75 2255  23.82
16 175 0.50 105 40.00 14.46 1498  18.69
17 54 0.30 75 24.00 24.28 2457 2345
18 195 0.30 75 24.00 23.89 23.80  24.89
19 125 0.02 75 24.00 24.49 25.08  24.09
20 125 0.58 75 24.00 26.73 2635 2517
21 125 0.30 32 24.00 18.53 20.74  16.69
22 125 0.30 117 24.00 23.04 21.03  23.65
23 125 0.30 75 1.37 16.74 16.98  15.06
24 125 0.30 75 46.63 17.02 16.99  17.38
25 125 0.30 75 24.00 23.53 2358 2372
26 125 0.30 75 24.00 24.53 2358  23.72
27 125 0.30 75 24.00 22.89 2358  23.72
28 125 0.30 75 24.00 22.53 2358  23.72
29 125 0.30 75 24.00 23.93 2358 2372
30 125 0.30 75 24.00 24.53 2358  23.72

Predicted Value = Pred. value, Study Type: Response Surface, Runs: 30, Initial Design: Central Composite, Design
Model: Quadratic.
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Figure 5. Parity plots; (A) Response Surface Methodology (RSM) predicted against experimental actual
values. (B) Artificial Neural Network (ANN) predicted against experimental actual values. (C) RSM
predicted values against ANN predicted values.

Table 8. Artificial neural network (ANN) optimum values for hidden layer sizes and corresponding
transfer functions (‘tansig’ and ‘logsig’) for minimum error fittings for train, validation, and test of

optimized RSM data.

Train Validation Test All

Hidden Neurons 2 2 2 2

R RMSE MAE R RMSE MAE R RMSE MAE R

4-8* 0.99 0.09 0.06 0.99 0.07 0.04 0.99 0.08 0.06 0.99

8-4 0.99 0.03 0.01 0.95 0.10 0.09 0.95 0.11 0.09 0.94

8-16 0.99 0.04 0.03 0.78 0.17 0.13 0.96 0.12 0.09 0.93
9-19 0.99 0.02 0.01 0.93 0.13 0.08 0.99 0.11 0.09 0.96
11-4 0.99 0.04 0.03 0.85 0.11 0.09 0.95 0.14 0.11 0.96
11-7 0.99 0.02 0.01 0.97 0.07 0.06 0.82 0.12 0.09 0.97
11-9 0.99 0.02 0.01 0.99 0.11 0.09 0.83 0.15 0.12 0.96
13-9 0.95 0.07 0.05 0.92 0.14 0.13 0.94 0.12 0.09 0.93
13-13 0.99 0.03 0.02 0.76 0.11 0.09 0.93 0.16 0.13 0.95
17-10 0.99 0.02 0.01 0.90 0.12 0.13 0.99 0.06 0.05 0.96
17-18 0.99 0.02 0.01 0.91 0.11 0.08 0.94 0.12 0.09 0.97
19-4 0.99 0.03 0.02 0.52 0.12 0.09 0.97 0.12 0.09 0.96
19-6 0.98 0.07 0.05 0.83 0.12 0.11 0.99 0.02 0.02 0.96

*4-8 is the chosen model based on its comparative high R? value.

As shown in Figure 5 below, the R? value is a good revelation of the compatibility of each data set
to each other.

The RSM values shows a high R? value of 0.9563 than the ANN R? of 0.944 with a negligible
residual value of 0.0123 between the RSM and ANN. Hence, the ANN is a very good method of
validating RSM data set [21]. The process of validating the Levenberg-Marquardt back propagation
model for the response yield of photoluminescent quantum yield adopted in this study were done
by deploying different adjustable topologies (see Table 8) in training of the network performance.
From the 13 topologies deployed hidden layers between 4 and 20; the best hidden layer configuration
with high coefficient of determination and low training error was gained at 4-8, as evident in Figure 6.
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Figure 6. Artificial neural network model coefficient of determination and error relationship of data set
of fluorescent carbon dots. (a) Trained R2 output, (b) Validated R2 output, (c) Test R2 output, and (d)
Overall R? output of data set.

Figure 6 below shows a high value of R? > 95 for the output parameter of photoluminescent
quantum yield which complies with the expected results from training of data set by the
Levenberg-Marquardt algorithm [17,29].

3.4. Characterization and Properties of Carbon Dots

The need to analyze carbon dots for their characteristic attributes is very essential. These can be
done by determining the particle sizes and morphological patterns using high resolution devices [34].
The atomic force microscopy (AFM), high resolution transmission electron microscopy (HrTEM), and
field emission scanning emission microscopic (FESEM) techniques have been utilized for this purposes.

3.4.1. Atomic Force Microscopy (AFM) and High Resolution Transmission Electron Microscopy
(HrTEM) of Carbon Dots (CDs)

CDs particle size distribution and morphology have been investigated (Figure 7). Figure 7a shows
the three dimensional (3D) plot of the morphological pattern of carbon dots while Figure 7b depicts 62
counts of CDs with a mean height and diameter of 4.054 nm and 44.032 nm, respectively, which is a
confirmation of the nano-dimension of CDs. Figure 7c represents the histogram of the 62 carbon dots
count and Figure 7d is the HrTEM of carbon dots at less than 10 nm and a lattice spacing at 0.24 nm.
HrTEM analysis for the carbon dots were investigated to determine the actual size and shape of the
carbon dots. The image clearly depicts the synthesized CDs as well dispersed in water with a spherical
petal shape and fine size distribution of about 3.0-5.0 nm in diameter shown in Figure 7.
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Figure 7. Atomic force microscopy (AFM) and high resolution transmission electron microscopy
(HrTEM) of carbon dots (CDs). (a) 3D CDs particles, (b) 2D CDs particles, (c) CDs particle distribution,
and (d) CDs particle sizes and lattice space.

The HrTEM provides a validation of the nano dimensions present in the synthesized CDs via
hydrothermal route which is in agreement with semiconductors synthesized at the nano scale [35,36].

The CDs lattice spacing of 0.24 nm renders it suitable in membrane filtration application. More
so, the sizes of CDs are below 5 nm, which means there are numerous surface sites for adsorption
application purposes in wastewater treatment.

In addition, Table 9 shows a detailed presentation of the size distribution of CDs obtained
by measuring the heights, areas, and diameters of 62-single carbon dots observed under atomic
force microscopy.

Table 9. Analysis of the atomic force microscopy of CDs.

Parameter Mean Minimum Maximum
Total Count 62 62 62
Height 4.054 (nm) 2.174 (nm) 8.486 (nm)
Area 2086.516 (nm?)  381.470 (nm?)  18,005.371 (nm?)
Diameter 44.032 (nm) 22.039 (nm) 151.411 (nm)
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As presented on Table 9. The CDs mean diameter of 44.034 nm and mean area of 2086.516 nm?
possess the attributes of a suitable adsorbent for adsorbing environmental pollutants [13].

3.4.2. Field Emission Scanning Electron Microscopy (FESEM) and EDx of Tapioca-Derived
Carbon Dots

From the conducted FESEM analysis, it is found that the actual shape of the CDs is in the form of
a flower shaped petals, spherical in nature, as can be seen in Figure 8a,b. The EDX study determined
the elemental compositions of carbon dots, and the results shows presence of C (31.64%), 0 (55.84%),
Na (10.99%), Si (1.45%), and K (0.079%) as in Figure 8c. The Na signal in EDX spectrum is due to the
sodium hydroxide constituent of carbon dots synthesis and silicon is as result of glass substrate for
drying carbon dots during the sample preparation for FESEM.

Spectrum 1
Element | Weight Atomic
% %
C 24.20 31.64
o]
(C) o] 56.89 55.84
Na 16.08 10.99
MNa Si 2.59 1.45
K
c K 0.23 0.09
Total 100.00
Si
e KK
T T T T T T T T T T T T T T T T T T T
o 1 2 3 4 5 B 7 8 9
Full Scale 20654 cts Cursor- -0.097 (0 cts) ke

Figure 8. FESEM at (a) 40 um (b) 10 um and (c) EDx of tapioca-derived carbon dots.
3.4.3. Properties of Carbon Dots (CDs)

The use of UV lamp to assess the quality of fluorescent carbon dots have been applied as a
source to obtain a blue/green color in the near visible region of color band group. The UV irradiations
absorbed by the carbon dots and excited through absorbing the energy leading to an electron excited
state. The molecules of carbon dots with extended Pi-electron provides the basics for the fluorescence
emission of carbon dots. The tapioca-derived carbon dot is a wavelength dependent photoluminescent
ionic solution in the visible range with a surface abundant with hydroxyl and carboxylic/carboxyl
moieties [13].

CDs indicated a strong optical absorption in the UV region (230-340 nm) with a tail extending to
the visible range as presented in Figure 9.
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Figure 9. Optical properties of CDs. (a) UV-visible absorption and emission spectra of as prepared CDs
dispersed in water. (b) Fluorescence emission spectra of carbon dots and reference material (Tapioca).

Absorption shoulders in the spectrum may be due to the 7-rt* (pi to pi star transition) of C=C
bonds or n-7* (n to pi star transition) of C=0 [37]. The uniqueness of CDs is the photoluminescence
emitted by it. Based on past study, it shows the dependency of intensity and wavelength emission
towards excitation wavelength [38]. This is due to the different size of particles and surface chemistry
and different emissive traps on CDs surface that can be related by the synthesis method.

The wavelength dependence behavior makes CDs possible to be applied in multi-color imaging
applications. It has been suggested that there are separate emissions by CDs core and surface states
whereby size, surface, and defects are responsible for the emission properties [13]. The color of CDs
most of the time is related to the surface groups which compares to particles size and normally CDs
show strong photoluminescence from blue to green wavelength. In terms of chemical properties,
different synthesis methods of CDs lead to different chemical structure, such as polymer chains, oxygen
based and amino-based groups [39].

The main challenge with carbon dots is the agglomeration of the particles due to strong particle
interactions. It can be postulated that the agglomeration of particles over time is the very reason that
these CDs emit green fluorescence since it had been left for 3 months. The resultant change in colour
due to agglomeration of carbon dot particle can be observed in Figure 10 (A and B) [40].

Figure 10. Tapioca powder as source of fluorescent carbon dots. (A) Carbon dots after synthesis
emitting blue fluorescence. (B) Carbon dots emitting green fluorescence after 3 months of synthesis.

It is well-known that there is a relationship between emission wavelength of quantum dots
and particle size, i.e., the smaller the particle size is, the shorter the emission wavelength [41]. It is
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reasonable to speculate that this law is also applicable to carbon dots. Strong green photoluminescence
offers unique superiority, because most of the current carbon dots emit blue fluorescence under UV
irradiation. This kind of carbon dots had been synthesized [42].

4. Conclusions

The concept of synthesizing CDs (at zero dimension) on an industrial scale requires automation
by first being able to predict the possible outcomes based upon the intended experimental factors.
Thus, this study has applied tapioca as the material for optimization to achieve the set objective of large
scale synthesis by means of prediction through a reliable approach of utilizing design of experiment
(DoE) for a response surface methodology (RSM), to optimize a facile and effective synthesis process of
fluorescent carbon dots from tapioca powder (starch) via hydrothermal synthesis route. The prediction
for optimized fluorescent carbon dots synthesis from RSM is in excellent agreement with artificial
neural network prediction by the Levenberg-Marquardt back propagation (LMBP) algorithm in terms
of R?, root mean square error and mean absolute error. Positive hidden layer sizes have resulted in the
ANN prediction of PLQY of fluorescent carbon dots at 26.25% and RSM predicted value of 27.38% at
R? values of 0.94 and 0.95, respectively. The best parameters values for the synthesis of carbon dots
were at 170 °C for 1 h 40 min with solvent ratio of 12 mL and dosage 0.1 g. These optimization and
prediction process have produced sustainable, efficient, and reliable fluorescent carbon dots, which is
energy saving in a manageable time, along with a decreased dosage with optimum quality output.

Moreso, to confirm the validity of carbon dots, characterization of surface morphology and
particles size carbon dots were conducted with favorable confirmations of the unique characteristics
and attributes of synthesized carbon dots by hydrothermal route.
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Abstract: Combinations of spherical vessels and pipes are frequently employed in industries.
Scholars have primarily studied gas explosions in closed vessels and pipes. However, knowledge of
combined spherical vessel and pipe systems is limited. Therefore, a flame acceleration simulator was
implemented with computational fluid dynamics software and was employed to conduct natural gas
explosions in three structures, including a single spherical vessel, a single spherical vessel with a pipe
connected to it, and a big spherical vessel connected to a small spherical vessel with a pipe. These
simulations reflected physical experiments conducted by at Nanjing Tech University. By changing
the sizes of vessels, lengths of pipes, and ignition positions in linked vessels, we obtained relevant
laws for the time, pressure, temperature, and concentrations of combustion products. Moreover,
the processes of natural gas explosions in different structures were obtained from simulation results.
Simulation results agreed strongly with corresponding experimental data, validating the reliability
of simulation.

Keywords: combination system; flame acceleration simulator (FLACS); pipe length; ignition position

1. Introduction

Flammable gases are employed extensively in the production of petrochemicals, and regularly
stored and transported using linked vessels. A typical linked system comprises closed vessels connected
with pipelines [1]. However, widespread fires and explosions frequently occur in such structures
because the flames and shock waves that result from local gas explosions can propagate through the
pipeline [2-4]. Although the prevention and mitigation techniques for fires and explosions have been
continuously improving, the number of accidents has not decreased in recent years. For instance,
an explosion and a fire accident caused by a gas leak considerably damaged two nearby buildings,
on 23 December 2008, in a coal gasification plant in Hunan province in China [5]. Furthermore,
an explosion occurred in 2013 in Dalian Bay, China, due to an operation that violated safety protocols
and ignited flammable gases in a vessel, which caused the death of two individuals and severely
injured two others. The explosion that occurred in natural gas pipes in 2017 in Guizhou Province,
China, caused the death of eight individuals and injured 35 individuals. Therefore, conducting further
studies on gas explosions occurring in linked vessels is of substantial importance for preventing and
mitigating the damage caused by gas explosions.
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Research on the structure of linked vessels has been conducted in past decades. Generally,
compared with methane explosions occurring in vented single vessels, gas explosions in linked vessel
systems constitute a complex process. This complexity can lead to high explosion strength in linked
vessels [6]. Furthermore, numerous influential factors, including venting sizes and positions, ignition
positions, and pipe lengths, could affect this process, among which ignition positions or obstacles could
considerably change flame propagation, accelerate the speed of fire, and rapidly increase explosion
pressures [7-12]. The venting size and position also play a prominent role in gas explosions [13-18].
Studies have demonstrated that the pressure piling exists in linked vessels, and the explosion strength
is principally affected by the pipe length to volume ratio [19-25].

Currently, along with the development of computer techniques, numerical simulations have
become more advanced. Numerical simulation has numerous advantages over physical testing;
numerically simulated experiments are cost-effective, user-friendly, and compatible with a limitless
number of experimental devices. Some scholars have examined gas explosions by using numerical
simulations. With experimental data collected from the literature and the computational fluid
dynamics (CFD) software AutoReaGas, Maremonti et al. [24] simulated gas explosions in two
linked vessels. They not only demonstrated that turbulence induced in the second vessel was
a major factor influencing violence of the explosion but also verified the validity of their CFD
code. Deng et al. [25] conducted an explosion experiment with a CHy and CO mixture in a 20-L
nearly-spherical tank and then used flame acceleration simulator (FLACS) software to mimic the gas
explosion of the experiment. They compared their simulation results with experimental data to prove
the reliability of the simulation. Ferrara et al. [26] modeled gas explosions vented through ducts by
using a two-dimensional (2D) axisymmetric CFD model based on the unsteady Reynolds-averaged
Navier—Stokes approach. Simulation results evidenced that the severity of ducted explosions is mainly
influenced by vigorous secondary explosions occurring in the duct. Valeria et al. [27] used a validated
large-eddy simulation model to study the mechanism underlying vented gas explosions in the presence
of obstacles. Methane—air mixtures with different composition ratios, variously shaped obstacles,
and area block ratios were investigated. The influences of the combustion rate and venting rate on
both the number and intensity of overpressure peaks were observed.

In this study, to systematically analyze how the vessel size, vessel structure, ignition position,
and length of connection pipes influence natural gas explosions, the strongly validated N-S solver
tool of FLACS [28], which has been developed continuously for more than 40 years for predicting the
consequences of gas explosions [29,30], was employed. This tool includes a three-dimensional CFD
code that solves Favre-averaged transport equations for mass, momentum, enthalpy, turbulent kinetic
energy, rate of dissipation of turbulent kinetic energy, mass-fraction of fuel, and mixture-fraction on
a structured Cartesian grid using a finite volume method. The Reynolds-averaged Navier-Stokes
equations are closed by invoking the ideal gas equation of state and the standard k—¢ model for
turbulence. Furthermore, one of the key features that distinguishes FLACS from most commercial
CFD codes is the use of the distributed porosity concept for representing complex geometries on
relatively coarse computational meshes [28]. With this approach, large objects and walls are represented
on-grid, whereas smaller objects are represented sub-grid. The pre-processor Porcalc reads the grid and
geometry files and assigns volume and area porosities to each rectangular grid cell. In the simulations,
the porosity field represents the local congestion and confinement, and this allows sub-grid objects to
contribute with flow resistance (drag), turbulence generation and flame folding in the simulations.

Simulated-pressure data were compared with the experimental results received from Nanjing Tech
University, where the explosion experiment with 10 vol% methane was conducted [13]. In addition,
results revealed the distributions of temperature and concentrations of gas products occurring
during gas explosions. This study mainly aimed to contribute to this research field by exploring the
characteristics of gas explosion in linked vessels.
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2. Simulation Objects

2.1. Experimental Apparatus and Geometric Model

The special-designed experimental device and its size and structure are displayed in Figure 1.
This experimental system consisted of two different spherical vessels, where one is big and the other is
small comparatively, and they connected each other with pipes having a varied length [13].

(b)

Figure 1. Picture and schematic diagram of the experimental device for the gas explosion. (a) Photo of
the experimental device; (b) scheme structural diagram of linked vessels.

Two spherical vessels were fabricated from steel with a design pressure of 20 MPa and their
internal diameters were 0.6 and 0.35 m (0.113 m3 and 0.022 m? in volume, respectively). Seamless
steel pipes, with an inner diameter and a thickness of 0.06 and 0.015 m, respectively, were connected
using flanges and bolts. The pipes were divided into three parts (each part was 2 m in length), so that
they could be combined into pipes with varied lengths of 2.45, 4.45, and 6.45 m between the big and
small spherical vessels. Suitable monitor points were placed as dictated by the experiment. The entire
explosion reactor with three different structures and the mesh are given in Figure 2.

Although only pressure data could be obtained from the experiment, the data regarding the
temperature and explosion products were determined through numerical simulation. It provided
further analysis and prediction dealing with unidentified explosion processes in linked vessels.

Based on the physical experiments, this study conducted natural gas explosions in three systems,
which were a single spherical vessel, a single spherical vessel connected with a pipe, and a big spherical
vessel connected to a small spherical vessel, individually. Meanwhile, multiple influence factors,
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including different structures, sizes of the vessels, lengths of the connection pipes, and ignition positions
were investigated in detail.
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Figure 2. Grid division on the model of the three structures. (a) Grid division on the model of the
spherical vessel; (b) grid division on the model of the single sphere with a pipe; (c) grid division on the

model of the big vessel connected with the small vessel.
2.2. Boundary Conditions and Initial Conditions

According to the experimental conditions and the FLACS User’s Manual [28], the initial conditions
were set as standard pressure and temperature and the boundary conditions were set as Euler equations
(inviscid flow equations), which is suitable for most explosion simulations. In FLACS, Euler equations
are discretized for a boundary element. Namely, the momentum and continuity equations are solved
on the boundary in the case of outflow. The concentration of methane used in the experiment was
10 vol%, and the vessel was filled entirely with the gas mixture. The parameter for the relative
fuel-oxygen concentration employed in FLACS is denoted by equivalence ratio (ER), which is defined
in Equation (1):
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ER = (mﬁml /moxygen)acmal — (VfUEI /Voxygen)actual (1)

(mfuel / moxygen) (Vfuel / Voxygen)

where m is the gas mass and V is the gas volume. Accordingly, the ER of methane-oxygen was 1.058.
In addition, the default choice of heat switch is close in FLACS, as large-scale explosions are not much
influenced. Activating radiation model will let walls and objects have background temperature, and
gas heat loss from radiation will calculate as well as radiation from hot objects around. This is useful
for small-scale confined explosions with important heat effects and is suitable to apply in this study.
Furthermore, based on the experiment results in small closed vessels, Luo et al. [31] confirmed that
adding a radiation model is an effective way to improve the precision of calculated results. They found
that the simulation results without adding the radiation model had a large error with an average
error of 10.05%, while the simulation results with adding the radiation model had a small error with
an average error of 1.88%. Therefore, adding a radiation model to the simulation of gas explosions
was able to produce a sound agreement with experimental results. As a result, to examine a realistic
situation, this study included a radiation model to improve the accuracy of the results.

stoichiometric stoichiometric

3. Simulation Results with Different Structures
3.1. Simulation Results in the Single Spherical Vessel

3.1.1. Influence of Vessel Size on the Explosion Pressure

The center of the large sphere was at (0.35, 0.35, 0.35); the sensor mounted on the sidewall was at
(0.075, 0.35, 0.35); the center of small sphere was at (0.19, 0.19, 0.19); and the sensor mounted on the
sidewall was at (0.025, 0.19, 0.19). The simulation was performed following the methane explosion
experiment at 10 vol% in two single vessels. The results of the simulation are plotted in Figure 3.

—— Small vessel
- - — Big vessel

-
el 1l 1 1 1 1 1 1

0
0.00 0.02 0.04 0.06 0.08 0.10 0.12 0.14 0.16 0.18 0.20
Time (s)

Figure 3. Natural gas explosion pressure varying with time in two single vessels.

The results revealed that the pressure plots at the center of the spheres and the sensor on the
sidewall could be superposed. As shown in Figure 3, a nearly consistent maximum explosion pressure
can be observed in the big/small vessel, which were 7.78 and 7.79 barg, respectively. However,
the maximum explosion pressure in two vessels peaked at 102.9 and 53.6 ms, respectively, and there
was a considerable difference of 49.3 ms, where the amplitude can be attained to 48%. Therefore, the
rate of pressure rise in the big vessel was lower than that of the small vessel. As a result, without
changing the initial conditions of the scenario, the peak pressure was same, but by contrast, the rate of
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the pressure rise was only relevant to the vessel size. Additionally, this finding was consistent with the
“explosion cubic law” [32].

3.1.2. Simulation Results of the Explosion Parameters in the Single Vessel

The distributions on the X-Y section of explosion parameters containing pressure, temperature,
and explosion products at the 10 vol% CH, mixture in the big vessel are presented in Figure 4.
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Figure 4. Contours of the natural gas explosion parameters containing pressure, temperature, and
concentration of products in the larger vessel. (a) Distribution of pressure; (b) distribution of temperature;
(c) distribution of the concentration of products.

As evidenced in Figure 4a, the pressure in the larger vessel was practically uniform at any specific
time. The explosion wave expanded into a spherical wave and the temperature was high near the
igniter but gradually decreased until it approached the leading edge of the combustion wave in
Figure 4b. Here, the combustion wave reached the sidewall at 125 ms. The temperature in the big
vessel began to decrease when the explosion ended. As indicated in Figure 4c, when more natural gas
participated in the reaction, the distribution of the explosion products spread further until the gas was
consumed. The laws of the distribution of pressure, temperature, and explosion products in the small
vessel were the same as those in the big vessel.
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The changing temperatures and concentrations of the explosion products in the vessels are plotted
in Figure 5. When the explosion occurred, the temperature at the center of the vessel enhanced rapidly,
reaching 2161 K at 6 ms. Then the rate of the temperature rise dropped, and the temperature at the
center of the small vessel reached 2580 K at 53 ms, whereas in the big vessel it reached 2581 K at 101 ms.
The center and sidewall reached the highest temperature simultaneously, but the energy was lost when
the combustion wave contacted the sidewall. The highest temperature near the sidewall was lower
than that at the center of the sphere. After it was ignited at the center of the sphere, the concentration
of products at the center escalated promptly, reaching its highest value (27.7%) at 12 ms in Figure 5b.
The theoretical concentration of products reached 28.35% based on the methane combustion chemical

equation. Accordingly, the simulation results fitted the theoretical values.
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Figure 5. Profiles of the natural gas explosion parameters in the single vessel. (a) Profiles of explosion

temperatures against time in the vessels; (b) concentration profiles against time for the explosion
products in the vessels.

3.2. Simulation Results in for Single Vessel with Pipes Connected to IT

3.2.1. Simulation Results of the Explosion Parameters in the Single Vessel with Pipes Connected to It

The big sphere with the 4.25-m pipe was adopted as an example; the igniter was located at the
center of the sphere. The distributions of the pressure, temperature, and concentration of products
of the methane explosion are shown in Figure 6. The plots of the explosion temperature and the
concentration of products are delineated in Figure 7.

Figure 6a shows that the peak pressure in the single vessel with a pipe connected to it was lower
than that of the single vessel. As provided by Figure 6b, because of the oscillation sparked by the blast,
the gas in the pipe was carried to the vessel, and a jet flow formed. Moreover, from Figure 6¢, when
the ignition source was located at the center of the vessel, the combustion wave spread 2.3 m in the
pipe when it spread from the center to the sidewall in the vessel. Therefore, compared with the vessels

alone, the pipes significantly accelerated the spread of the explosion.

In virtue of the oscillation of the explosion, the plots of temperature displayed a fluctuation at the
center of the vessel (in Figure 7a). Moreover, the energy was lost when the combustion wave contacted
the sidewall. The highest temperature near the sidewall was low and the fluctuations were small
comparatively. The plots in Figure 7b reveal that as the explosion wave spread, the entire vessel was
filled with combustion products, which reflected that the variation of products was not be affected by

the oscillation.
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Figure 6. Contours of the natural gas explosion parameters (pressure, temperature, and concentration
of products) in the single vessel with a pipe connected to it; (a) distribution of pressure. (b) distribution

of temperature; (c) distribution of the concentration of products.
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Figure 7. Profiles of the natural gas explosion parameters (temperature and concentration of products)
in the single vessel with a pipe connected to it.

3.2.2. Influence of Changing the Ignition Position

The big vessel with the 4.25-m pipe was adopted as an example, the gas was ignited at the sidewall
of the big vessel, the center of the sphere, location 1, and location 2 (in Figure 2). The plots of the
explosion pressure as it changed with time are illustrated in Figure 8.
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Figure 8. Natural gas explosion pressure versus time by changing the ignition positions.

As revealed in Figure 8, changing the ignition position substantially influenced the explosion
pressure in the big vessel. The peak pressure locations in the big vessel with different ignition positions
could be sorted from highest to lowest as follows: location 4, location 2, the center, location 1, and the
sidewall, orderly. By contrast, the maximum rate of pressure rise could be sorted as follows: location
4, location 2, location 1, the center, and the sidewall, orderly. The combustion wave contacted the
sidewall during the explosion when the ignition source was located at the sidewall. Due to the wall
effect, the abundant energy was consumed, which resulted in the explosion intensity to be weaker than
at the other ignition positions. Accordingly, the explosion pressure and the rate of pressure rise were
diminished. Locations 1, 2, and 4 were located within the pipe, where the explosion occurred with
a tighter constraint and the pipes accelerated the explosion wave. Location 4 was at the end of the
pipe, at which the acceleration distance was the longest. The explosion intensity was most vigorous
when the combustion wave spread to the big vessel. Therefore, the explosion pressure and the rate of
pressure rise were the largest when the ignition source was at location 4.
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3.2.3. Influence of the Length of the Connection Pipes

The ignition source was located at the center of the vessel. The explosion pressure in different
lengths of pipe is shown in Figure 9.
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Figure 9. Natural gas explosion pressure versus time by changing the length of the pipe connected to
the single vessel.

Figure 9 shows that the size effect [33] of each pipe length had a noticeable influence on the
explosion intensity in the vessel. The peak explosion pressure lessened as increasing length of the
connection pipe, with the influence being the greatest in the small vessel. In any single vessel with
a connected pipe, the energy from the explosion was consumed by the wall. More energy was
lost when the lengths of the connected pipes increased on account of the greater area of the inner
wall. Consequently, the peak explosion pressure decreased with the increase in the lengths of the
connected pipes.

3.3. Simulation Results of the Big Vessel Connected to the Small Vessel

3.3.1. Simulation Results of the Explosion Parameters in the Big Vessel Connected to the Small Vessel

Regarding the simulation of the big vessel with the 4.45-m pipe and the small vessel, the ignition
source was located at the center of the big vessel. Changes in the distribution of the pressure,
temperature, and concentration of products are presented in Figure 10. The plots of the temperature
and the concentration profiles of the explosion products are depicted in Figure 11.

Above

25 ms 7.5
. 7.0

75 ms 5.5

125 ms ) 3.5

175 ms

(@)

Figure 10. Cont.

336



Processes 2020, 8, 52

-

160
.2400
2200

o4 25 ms
02 . 2000
0. 75 ms 1800
04 C 1600
02
1400
0.6) )
0,4 125 ms . 1200
0.24% 1000
0. 175 ms 800
0.4)
b %\,) 600
m
1.0 1.5 20 25 30 35 40 45 5.0 00
(b)
.5
UA‘{(m) . .0.26
04 23115 . 024
02 022
ad 0.20
. 7
04 5 ms . 0.18
by 0.16
0.14
z:’ 125 ms 0.12
o R
0.6 175 ms 0.06
0.4 , 0.04
0.2] 0.02
0.5 1.0 15 20 25 30 35 40 45 350 w 0.00
(0)

Figure 10. Contours of the natural gas explosion parameters in the big vessel connected to the small
vessel. (a) Distribution of pressure; (b) distribution of temperature; (c) distribution of the concentration
of products.
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Figure 11. Profiles of the natural gas explosion parameters (temperature and concentration of products
versus time) in the big vessel connected to the small vessel. (a) Profiles of explosion temperatures against
time in the vessels; (b) concentration profiles against time for the explosion products in the vessels.

As illustrated in Figure 10a, the peak pressure was attained in the small vessel at 75 ms and
was considerably higher than that in the big vessel. The temperature in the pipe was lower than
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that in the sphere because energy was absorbed by the wall in the long pipe in Figure 12b. Gas at
a lower temperature was brought into the sphere as a result of the oscillation originated from the
explosion wave. Here, the low-temperature jet cuts the temperature distribution in the sphere to a
symmetrical structure.
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Figure 12. Natural gas explosion pressure versus time in the large and small vessels at different
ignition positions.

In Figure 11a, the explosion wave reached its highest temperature (2700 K) at 70 ms in the small
vessel. By contrast, the temperature was 2400 K in the big vessel at the same time. Because of the
oscillation of the combustion wave, the temperature at the center of the big vessel plummeted to 1300 K
at 74 ms after the combustion wave spread into the small vessel, which then formed a second peak.

Figure 11b shows that the concentration of products enhances rapidly at 62 ms. Namely,
the combustion wave spread into the small vessel through the connected pipe at that moment.
Furthermore, the combustion wave spread to the sidewall of the big vessel at 65 ms. This result again
demonstrated the fact that the pipes accelerated the combustion wave.

3.3.2. How Changing the Ignition Position Influenced the Gas Explosion

Considering the big vessel with the 4.45-m pipe and small vessel as an example, the ignition
positions were the sidewall of the big vessel, the center of the big vessel, location 1, location 2, location
4, location 6, the center of the small vessel, and the sidewall of the small vessel, as depicted in Figure 2.
A methane mixture with a concentration of 10 vol% was intentionally selected in the simulations.
The plots of the explosion pressure in the large and small vessels are shown in Figure 12.

As exhibited in Figure 12a, the peak pressure was attained in the big vessel when the ignition
was positioned at the sidewall of the small vessel. The minimum pressure value was obtained when
the ignition was positioned at the sidewall of the big vessel. Figure 12b shows that the peak pressure
decreased when the ignition was moved from the sidewall in the big vessel to location 2, then the
peak pressure increased slightly when the ignition was moved from location 2 to the sidewall in the
small vessel. Therefore, the pressure piling [21,22] originated from the oscillation was the weakest at
location 2.

3.3.3. How Changing the Length of the Connection Pipe Influenced the Gas Explosion

In the experiment, the vessel that contained the ignition source was defined as the initiating vessel,
and the other was defined as the secondary vessel. The explosion results were determined variably
with the change in the length of the pipe, as drawn in Figure 13.

Figure 13a,b exhibited that the length of the pipe affected the pressure in the initiating vessel.
The peak pressure in the initiating vessel was reached when the length of the connecting pipe was
4.45 m. As the length increased, the peak pressure in the initiating vessel increased, yet decreased when
the pipe was longer than 4.45 m. The acceleration from the pipes on the combustion wave reinforced
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the explosion intensity, but as the length of the pipe increased, more energy was dissipated by the cause
of the wall. When the length of the connecting pipe was excessive, a more momentous influence on the
acceleration from the wall than that from the pipes could be demonstrated. Comparing Figure 13a
together with Figure 13¢,b together with Figure 13d revealed that the peak pressure in the initiating
vessel was lower than that in the secondary vessel. Moreover, when the initiating vessel was the big
vessel, the peak pressure in the second vessel was considerably higher.
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Figure 13. Natural gas explosion pressure-time curves in the vessels with changing the length of
the pipe.

4. Error Analysis

The simulation results and experimental data were compared for two different structures, which
were the big vessel connected to the small vessel with the 4.45-m pipe and the 6.45-m pipe. The peak
pressure values at different ignition positions from the simulation and experiment are shown in
Figure 14.

In Figure 14a, “Al1” to “A7” correspond to the adopted ignition positions: the sidewall of the big
vessel, the center of the big vessel, location 1, location 2, location 4, the center of small vessel, and the
sidewall of the small vessel. Similarly, in Figure 14b, “B1” to “B8” represent the adopted ignition
positions: the sidewall of the big vessel, the center of the big vessel, location 1, location 2, location 4,
location 6, the center of the small vessel, and the sidewall of the small vessel.

Figure 14 shows that the variation that depended with the ignition positions of the peak pressure
in the simulation and experiment were analogous. From Figure 14a, the average error between
the simulation and experimental data is 7.7% in the big vessel with the 4.45-m pipe connected to it.
The average error is shown in Figure 14b is 8.04%. Overall, 70% of the simulation data were marginally
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larger than those of the experimental data in all the compared points, in which the average difference
was 6.97%. Generally, an experiment was controlled by more factors in comparison with a simulation.
Even though the radiation model was employed, with the aim of improving simulated accuracy, some
degree of discrepancy still could be observed. The increase of the temperature in the wall of the pipes
was limited in the physical experiment. A considerable temperature difference existed between the
combustion products and the wall of the pipe. The energy was dissipated on account of thermal
convection, thermal conduction, and thermal radiation. As a result, the simulation data had higher
values than the experimental data [34]. Consequently, on the one hand, the simulation effectively
reproduced the development of the gas explosion in linked vessel. On the other hand, it correctly
predicted the variations of pressure, temperature, and the changing concentration of products with
different factors.

20 2.0,
[Fe—Peak pressure in large sphere by simulation 18- [—®—Peak pressure in large sphere by simulation
L8 |—e—Peak pressure in large sphere by experiment!
L6k |-a—Peak pressure in small sphere by simulation
1.6 - : |-v—Peak pressure in small sphere by experimen]
14 1.4+
s 12t g 121
% 1.0 - < 10F
§ «
afosh oF osf
0.6 - 0.6
04 . 04 .
(a) Big vessel connected to the (b) Big vessel connected to the
02 | small vessel with a 4.45-m pipe 0.2 small vessel with a 6.45-m pipe
0.0 L L L L L L L 0.0 L L L L L L L L
Al A2 A3 A4 A5 A6 A7 : BI B2 B3 B4 B5 B6 B7 B8
Ignition position Ignition position

Figure 14. Comparison between the simulation data and experimental data of the natural gas explosion
pressure versus time in the linked vessels.

5. Conclusions

In this study, natural gas explosions were conducted, employing the simulation tool of FLACS in
three systems, including a single vessel, a single vessel with a pipe connected to it, and a big vessel
connected to a small vessel with a pipe. The main conclusions of this study are as follows:

e  When the explosions occurred in the same initial conditions, the peak explosion pressure
maintained unchanged. However, the maximum rate of pressure rise decreased when the volume
of the closed vessels increased, which is following the explosion cubic law.

e  The length of the connecting pipe profoundly affected the explosion intensity in the single vessel
with a pipe connected to it and in the big vessel connected to the small vessel. Furthermore,
an oscillation of the explosion wave developed in the linked vessels.

e  Theignition position affected the explosion intensity in the linked system extremely. The explosion
intensity was weakest when the ignition source was positioned at the center of the system.

e When the explosions occurred in the linked system, the explosion intensities in the secondary
vessels were stronger than those in the initiating vessels.

e  Using FLACS software for the simulation of the small-scale linked system, the average difference
attained 6.97%. Accordingly, the simulation results agreed reasonably well with the experimental
results and the actual explosion conditions were reflected in the FLACS simulation to an
acceptable extent.
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